Author’s response to reviews

Title: Exploring strategies to reach individuals of Turkish and Moroccan origin for health checks and lifestyle advice: a mixed-methods study

Authors:

Andrea Bukman (sandra.bukman@wur.nl)
Dorit Teuscher (d.teuscher@maastrichtuniversity.nl)
Jamila Ben Meftah (j.ben_meftah@lumc.nl)
Iris Groenenberg (i.groenenberg@lumc.nl)
Mathilde Crone (m.r.crone@lumc.nl)
Sandra van Dijk (s.van_dijk@lumc.nl)
Marieke Bos (m.bos@hartstichting.nl)
Edith Feskens (edith.feskens@wur.nl)

Version: 2 Date: 17 Jun 2016

Author’s response to reviews:

Dear editorial board of BMC Family Practice,

Thank you for providing us with the opportunity to revise our manuscript, entitled:

‘Exploring strategies to reach individuals of Turkish and Moroccan origin for health checks and lifestyle advice: a mixed-methods study’. Hereby we send you our revised manuscript.

The only comment left, was from the editor:

“Please provide more detailed and accurate information about the quantitative phase of your mixed-methods study.”
The comment helped us to critically reconsider our description of the quantitative phase of our mixed-methods study. To provide a more detailed description of the methods, we added information regarding the questionnaire administration period, questionnaire duration, number of responders on the web-based questionnaire, and ethnic origin of the subjects of the pilot test (lines 138–155). To provide more detailed results, we added p-values of Chi-Square tests comparing the answers of participants of Turkish and Moroccan origin (lines 259–260, lines 407–408, Table 4 and Table 5).

We hope that the revised manuscript is now suitable for publication in BMC Family Practice. We look forward to hearing from you.

Kind regards,

On behalf of all authors

A.J. Bukman, MSc.